Author’s response to reviews

Title: Craniosacral Therapy for chronic pain: a systematic review and meta-analysis of randomized controlled trials

Authors:
Heidemarie Haller (h.haller@kem-med.com)
Romy Lauche (Romy.Lauche@sozialstiftung-bamberg.de)
Tobias Sundberg (tobias.sundberg@ki.se)
Gusatv Dobos (g.dobos@kem-med.com)
Holger Cramer (h.cramer@kem-med.com)

Version: 2 Date: 18 Nov 2019

Author’s response to reviews:

Thank you for the opportunity to revise our manuscript.

As suggested, we added a conclusions section below the discussion. Sorry for the delay.

Kind regards.

On behalf of the authors.
Heidemarie Haller